Author's response to reviews

Title: Indivo: a personally controlled health record for health information exchange and communication

Authors:
Kenneth D Mandl (Kenneth_Mandl@Harvard.edu)
William W Simons (william.simons@childrens.harvard.edu)
William CR Crawford (William.Crawford@childrens.harvard.edu)
Jonathan M Abbett (Jonathan.Abbett@childrens.harvard.edu)

Version: 5 Date: 6 September 2007

Author's response to reviews:
I have made the formatting changes requested.